Author's response to reviews

Title: Regulation of podocalyxin expression in the kidney of streptozotocin-induced diabetic rats with Chinese herbs (Yishen capsule)

Authors:

Jingai Fang (fja3455001@126.com)
Hongkun Wei (slyhwhk@163.com)
Yanyan Sun (yanyansun6839@126.com)
Xiaodong Zhang (zxdsp@163.com)
Wenyuan Liu (liuwenyuan3057@sina.com)
Qintao Chang (xhldstjy@163.com)
Ruihua Wang (wrh8807@163.com)
Yuewen Gong (Yuewen.Gong@ad.umanitoba.ca)

Version: 2 Date: 15 October 2012

Author's response to reviews:

Dear Carmela S. Quidoles

Thank you very much for providing us the opportunity to modify the manuscript. I have added the Authors’ Contributions section and Competing interest section in the revised manuscript. I hope that with this revision, the manuscript is suitable to go through review process in your journal.

Yours sincerely,
Yuewen Gong, MB, PhD
Professor of University of Manitoba